Author's response to reviews

Title: Prevalence and Risk Indicators of Gingivitis and Periodontitis in a Multi-Centre Study in North Jordan: a cross sectional study.

Authors:

Khansa T Ababneh (ktababneh@hotmail.com)
Zafer MF Abu Hwaij (dr_zafer79@yahoo.com)
Yousef S Khader (yousef.k@excite.com)

Version: 3 Date: 28 December 2011

Author's response to reviews:

Dear Editor of BMC Oral Health,

Thank you very much for accepting our manuscript entitled: “Prevalence and Risk Indicators of Gingivitis and Periodontitis in a Multi-Centre Study in North Jordan: a cross sectional study”, MS ID : 1286668785398943, for publication in your reputed Journal. We have now made the necessary major and minor revisions to the format of the manuscript. However, I was unable to change the style of the author names in the references list, as I am out of my country at the moment and I do not have any access to the original journals to get the first name of each author; the names in Pubmed are written by surname.

Thanking you,
Sincerely,
Dr Khansa Ababneh